Author’s response to reviews

Title: Blood pressure and hypertension prevalence among oldest-old in China for 16 year: based on CLHLS

Authors:
Jiaju Du (yaoyaolunwen@sina.com)
Guoning Zhu (baoxuebo@163.com)
Yanhong Yue (jhwang301@163.com)
Miao Liu (liumiaolmbxb@163.com)
Yao He (380753512@qq.com)

Version: 4 Date: 26 Aug 2019

Author’s response to reviews:

Aug 27, 2019
Re: BGTC-D-19-00304R3

" Blood pressure and hypertension prevalence among oldest-old in China for 16 year: based on CLHLS”

Dear Editor:

Thank you very much for your positive comments on our manuscript. Our revised paper has incorporated all the suggestions. Below is our line by line response. We hope that the revised paper meets your approval for publication in BMC Geriatrics.

Sincerely
Miao Liu, Yao He

Institute of Geriatrics,
Chinese PLA General Hospital,
28 Fuxing Road, Beijing 100853, China.
Editor Comments:

1. Please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.

Reply: Thanks for your careful review. We have added accordingly (Conclusion part, line 23, page9).

2. In the Authors' contributions section, please refer to all authors by their initials rather than full names.

Reply: Thanks for your careful review. We have corrected accordingly (line3-8, page12).

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Reply: Thanks for your careful review. We have uploaded the clean version of the manuscript.

Once again, we really appreciate the comments and suggestions, which are valuable in improving the quality of our manuscript.

Should you have any questions, please contact us without hesitate.

With kindest regards,

Sincerely yours,

Miao Liu